Author's response to reviews

Title: Access to Primary Care is Associated with Better Autoimmune Hepatitis Outcomes in an Urban County Hospital

Authors:

Daniel Kim (dannerkim@gmail.com)
Daniel Eshtiaghpour (danesh0123@gmail.com)
Hal F Yee (hyee@dhs.lacounty.gov)
Anuj Datta (anujdatta@hotmail.com)
Viktor E Eysselein (veysselein@labiomed.org)
Joel Alpern (joelalpern@yahoo.com)

Version: 2  Date: 20 February 2015

Author's response to reviews:

Title content: An entire page was dedicated to the title page, and emails of the authors were added.

Abstracts: The "Background and Aims" headline was changed to "Background" only. It appears this was the only change needed.

Line numbering: Line numbers and page numbers were added.

Ethic statement: Ethic statement, standards of reporting, and data availability statements were added.

A competing interests statement was added.

Authors' contributions: A statement was added with each authors' contributions.

Acknowledgements: There were none to be added.